\input{ThisWeek.tex}
\newcommand{\duedate}{Thursday, September 11}
\input{../TemplateWeek/CheckHomeWork.tex}
\input{Exercises.tex}

%\booltrue{teacher}

\begin{enumerate}
\item The answer is the same as the next two
\item The answer is the same as the previous and the next
\item The answer is the same as the previous two
\item This really requires an argument in which you apply the Sandwich theorem, including identifying the two functions which bound $x^3\cos\frac{2}{x}$ from above and below, and identifying the limits of those two functions.  The answer should be $0$.
\item This requires an argument involving IVT.
\end{enumerate}

\end{document}